Author's response to reviews

Title: The co-administration of quetiapine or placebo to cognitive-behavior therapy in treatment refractory depression; a preliminary trial

Authors:

Yves Chaput (yveschaput@bellnet.ca)
Annick Magnan (annickmagnan@videotron.ca)
Alain Gendron (alaingendron@astrazeneca.ca)

Version: 10 Date: 21 August 2008

Author's response to reviews: see over
To the Editorial Staff,

BMC – Psychiatry,

MS: 8312126711678731

Montreal, August 21st, 2008.

I am sending you the last and final corrections for the manuscript entitled ‘The co-administration of quetiapine or placebo to cognitive-behavior therapy in treatment refractory depression; a preliminary trial’.

I have simply corrected two remaining typos and partially rephrased one phrase in the results section, where I have just kept ‘week 10’ rather and eliminated the redundant ‘day’ reference when referencing the patient withdrawn for the erroneous electrocardiogram. I have also added an ‘a’ to one of the phrases in the discussions section.

As you can see these changes are rather insignificant and I could probably go several other weeks putting ‘a’ in some phrases but at one point in time (the present time) this article has to be published. So I consider that this version ‘ChaputTRD21Aug08.doc’, unless something dramatic happens in the uploading process, needs no further corrections.

In the hope that this revision will meet with your satisfaction,

________________________
Yves Chaput, MD, FRCP(C), PhD